Author's response to reviews

Title: Tests for predicting complications of pre-eclampsia: A protocol for systematic reviews

Authors:

Shakila Thangaratinam (s.thangaratinam@bham.ac.uk)
Arri Coomarasamy (arricoomar@blueyonder.co.uk)
Steve Sharp (steve_sharp12@hotmail.com)
Khaled MK Ismail (khaled.ismail@uhns.nhs.uk)
Khalid S Khan (k.s.khan@bham.ac.uk)

Version: 5 Date: 1 August 2008

Author's response to reviews: see over
29\textsuperscript{th} July 2008

To
Editor
BioMed Central

Dear Editor

\textbf{Re: Tests for predicting complications of pre-eclampsia: A protocol for systematic reviews}

We have herewith attached the revised version of our manuscript. We would like to thank you for provisionally accepting our paper for publication.

Please do not hesitate to contact me if you have any further queries.

Yours Sincerely
Dr Shakila Thangaratinam MRCOG
Clinical Lecturer in Obstetrics and Gynaecology and Clinical Epidemiolgy
Academic Unit
3rd floor
Birmingham Women's Hospital
Birmingham B15 2TG, UK
E Mail: s.thangaratinam@bham.ac.uk
   shakila@doctors.org.uk
Ph: 07887775891 Fax: +44 (0)121 4154837